Author's response to reviews

Title: Population Attributable Fraction of Smoking and Metabolic Syndrome on Cardiovascular Disease Mortality in Japan: a 15-Year Follow Up of NIPPON DATA90.

Authors:

Naoyuki Takashima (takasima@belle.shiga-med.ac.jp)
Katsuyuki Miura (miura@belle.shiga-med.ac.jp)
Atsushi Hozawa (ahozawa@med.id.yamagata-u.ac.jp)
Aya Kadota (ayakd@belle.shiga-med.ac.jp)
Tomonori Okamura (okamurat@hsp.ncvc.go.jp)
Yasuyuki Nakamura (nakamura@kyoto-wu.ac.jp)
Takehito Hayakawa (thayaka@fuji.shiga-med.ac.jp)
Nagako Okuda (okuda@belle.shiga-med.ac.jp)
Akira Fujiyoshi (afujii@belle.shiga-med.ac.jp)
Shin-ya Nagasawa (nagasawa@belle.shiga-med.ac.jp)
Takashi Kadowaki (kadowaki@belle.shiga-med.ac.jp)
Yoshitaka Murakami (ymura@belle.shiga-med.ac.jp)
Yoshikuni Kita (kita@belle.shiga-med.ac.jp)
Akira Okayama (aokayama@jatahq.org)
Hirotugu Ueshima (huieshima@belle.shiga-med.ac.jp)
NIPPON DATA90 Research Group [] (huieshima@belle.shiga-med.ac.jp)

Version: 5 Date: 1 June 2010

Author's response to reviews: see over
To the Editor:

We have revised the manuscript according to the comments. Points of the revisions are summarized as follows:

**Major revisions (we require the author to make these changes)**

**Author group:** Please provide a list of the members of the author group in the authors acknowledgements section.

We added the list of the members in the acknowledge section.

**Authors’ contributions:** Please indicate that all authors read and approved the final manuscript.

We indicated that all authors read and approved the final manuscript in Authors’ contributions section.

**References:** Please make sure references are cited in numerical order. We notice reference 23 is not cited within the main text. Please make sure references continue to be cited in numerical order once this reference is cited.

We checked the manuscripts and included reference 23 within the main text.